FEATURED STUDENT RESEARCH PAPER



Contrasting aspects of tailwinds and asymmetrical response to crosswinds in soaring migrants

Paolo Becciu^{1,2} · Michele Panuccio^{2,3,4} · Carlo Catoni² · Giacomo Dell'Omo² · Nir Sapir¹

Received: 29 August 2017 / Revised: 16 January 2018 / Accepted: 22 January 2018 © Springer-Verlag GmbH Germany, part of Springer Nature 2018

Abstract

Billions of migrating birds travel between their breeding and over-wintering areas twice a year, encountering various environmental conditions along their migration routes. Wind is of utmost importance for birds as wind speed and direction may strongly affect the birds' flight speed and metabolism. Specifically, tailwinds were found to initiate flight and facilitate higher migration intensity and faster migratory movement while lowering the energetic cost of flight. Using radar, field observations and local meteorological measurements, we studied spring migrating raptors in a migration bottleneck in Italy near the Strait of Messina, between Sicily and Calabria. We explored the effects of wind on flight speed and the intensity of migration in soaring migrants. We found that the birds' ground speed was positively affected by tailwind speed, and thus, tailwind likely allowed the birds to reach their destination faster. In addition, bird airspeed decreased under increasing tailwind speed, presumably lowering the energetic cost of flight. These findings are in line with predictions of optimal migration flight theory, yet, tailwind had an unexpected negative effect on migration intensity. We suggest that tailwind conditions induced a change in route selection by the migrants at a regional scale, causing a local decline in migration intensity. This change involves the undertaking of cross sea flight rather than overland detour. Furthermore, we found a modular response to crosswinds as birds compensated for winds blowing towards the sea and drifted when winds blew towards land. Our findings suggest that migrating raptors respond to en route wind conditions and coastline geography by adjusting several features of their flight in a manner that will increase their travel speed, reduce the energetic cost of flight, and permit a safe journey.

Significance statement

Soaring birds exploit tailwinds to move faster and presumably to reduce the energetic cost of flight during migration. Using radar, we tested how bird flight is affected by tailwind and further predicted that the due to the advantages of using tailwinds, as increased number of birds migrating over the study area. We found that radar-tracked migrating birds flew faster with tailwinds over the study area, but despite this advantage, surprisingly low number of birds migrated when tailwinds prevailed. Furthermore, birds seem to drift with winds blowing from the sea to the land, but not in the opposite direction, suggesting that birds try to avoid drifting over the sea where it is riskier to fly. Hence, our study highlights the complexity of migrant flight behavior in relation to the wind and suggests a flexible response to different wind conditions and coastline geography at multiple spatiotemporal scales.

Keywords Bird migration · Environmental effects · Flight behavior · Honey buzzard · Wind speed · Radar ornithology

Communicated by W. Wiltschko

Electronic supplementary material The online version of this article (https://doi.org/10.1007/s00265-018-2447-0) contains supplementary material, which is available to authorized users.

Paolo Becciu pbecciu89@gmail.com

- ¹ Animal Flight Laboratory, Department of Evolutionary and Environmental Biology, University of Haifa, Aba Khoushy 199, 3498838 Haifa, Israel
- ² Ornis italica, Rome, Italy
- ³ MEDRAPTORS (Mediterranean Raptor Migration Network), Rome, Italy
- ⁴ DSTA, Department of Earth and Environmental Sciences, University of Pavia, via Ferrata 9, 27100 Pavia, Italy

Introduction

Environmental conditions are paramount in shaping migration properties in various organisms and may strongly affect the fitness of migrants (Boano et al. 2010; Fayet et al. 2015; Mellone et al. 2015; Rotics et al. 2016). Fitness-related costs caused by environmental conditions during long-distance migration have consequently modulated the evolution of migration schedules, flight behaviors, and migration route selection (Åkesson and Hedenström 2000; Ma et al. 2011; Bohrer et al. 2012; Flack et al. 2016). A central aspect of flight behavior is how an organism responds to varying atmospheric conditions and geographic features throughout its long-distance journey, and whether volant organisms modulate their response to atmospheric conditions based on their geographic position (Horton et al. 2016).

Among atmospheric conditions, wind is known to strongly influence the speed of migration under different ecological scenarios, and specifically during long-distance migration (Alerstam 1979; Liechti 2006; Mellone et al. 2012; Safi et al. 2013; Rotics et al. 2016; Vidal-Mateo et al. 2016). Because winds are constantly changing in speed and direction, flying animals must continuously adjust their flight behavior to efficiently move and accomplish their migration (Klaassen et al. 2011; Vansteelant et al. 2017b). Moreover, due to the benefits of flying with favorable wind, wind assistance may induce a decision to depart and hence may facilitate a high migration traffic rate (Åkesson and Hedenström 2000; Green et al. 2002; Klaassen et al. 2004) during day and night (Richardson 1978; Gwinner 1990). Several studies have shown that both flapping and soaring migrants travel significantly faster when flying with tailwinds and are hindered by headwind and crosswinds (Spaar and Bruderer 1996, 1997; Malmiga et al. 2014; Vansteelant et al. 2015). Also, bird airspeed decreased under tailwinds (Pennycuick 1978; Alerstam 1979; Liechti 1995). The reduction of airspeed under tailwinds reduces height losses during the gliding phase of flight in soaring migrants and allows birds to cover larger distances in a short time by reducing the time-consuming thermal circling phase of the flight (Spaar and Bruderer 1996). Moreover, reducing airspeed during the gliding phase may increase the overall time spent airborne and may also decrease the risk of reaching the ground or switching to energy-expensive flapping flight (Horvitz et al. 2014; Harel et al. 2016).

Crosswinds can have variable effects on the ground speed of different flying species. For example, in commuting fruit bats, the ground speed remained nearly constant with increasing crosswind speed (Sapir et al. 2014a), while for nine migratory waterfowl species, ground speed decreased with increasing crosswind speed, with the exception of one species that exhibited an opposite trend (Safi et al. 2013). Notably, soaring migrants travel more slowly under crosswinds than under tailwinds (Spaar and Bruderer 1996, 1997; Vansteelant et al. 2015). In fact, when flying at maximum range airspeed, the airspeed at which energy expenditure per unit distance traveled is minimized (Pennycuick 1978), a bird should reduce its airspeed in tailwinds and increase its airspeed in headwind and crosswinds, depending on the extent to which it compensates for wind drift (Liechti et al. 1994; Sapir et al. 2014b). Crosswinds may induce high energy costs due to the need to compensate for lateral drift and may further cause time losses due to the time it may take birds to return to their intended migratory path. In some circumstances, for example when birds drift away from land towards the sea, crosswinds may even cause mortality due to exhaustion (Thorup et al. 2003).

The sea-land interface and coastlines geography are therefore particularly relevant to long-distance migrants that pass over or near large water bodies. Alerstam and Pettersson (1977) reviewed studies documenting sea crossing, concentration of migrants near seashores, and flight directions of avian migrants along coastlines. These authors suggested that the birds decided whether to stop, follow the coast, or cross the sea based on considerations that take into account their tendency to drift over the sea, but not over land. Bird decisions were found to relate to the geographic settings (e.g., the width of the crossing and coastline direction in relation to goal direction), as well as the specific wind conditions at the crossing point. Specifically, soaring migrants prefer flying while exploiting strong thermals in order to attain low metabolism during their travel (Sapir et al. 2010), and since thermals are weak or absent over large water bodies, many soaring migrants try to avoid the crossing of oceans and seas. They consequently tend to take long detours over land (Meyer et al. 2000; Alerstam 2001; Panuccio et al. 2012; Nourani et al. 2016), concentrating in peninsulas, isthmuses, and narrow land corridors (Kjellén and Roos 2000; Bensusan et al. 2007; Bildstein et al. 2009; Nilsson et al. 2014; La Sorte et al. 2016).

One of the most important places where raptors migrate over land corridors is southern Italy at the center of the Mediterranean Sea where tens of thousands of raptors travel each spring after crossing the sea from North Africa. When the birds travel further north to central Italy, many of them pass via the Strait of Messina (e.g., Table S1 in Supplementary Materials) which is located between Sicily and Calabria (Fig. 1) (Panuccio 2011; Panuccio et al. 2016b). During their travel, the birds may encounter various wind conditions, some of which may cause them to drift over the sea. To avoid oversea flight, the birds may compensate for wind drift when the direction of the wind drifts them beyond the coastline, while no such compensation may take place where the direction of the expected drift is over the mainland. Yet, no such response has been so far documented in soaring migrants. Thus, the study area is particularly suitable to examine the response of soaring migrants to the combination of specific



Fig. 1 A map of the study area in Calabria, Italy, with radar tracks of migrating soaring birds on the 14th of May 2014, overlaid on a digital elevation model (DEM) with a resolution of 10 m as provided by "Istituto

meteorological conditions with the geographic features of the study area, potentially uncovering how soaring migrants may modulate their response to the wind according to the expected outcome of their drift. To quantify bird movement in a systematic manner, we used a radar that recorded key migration parameters that are otherwise difficult to obtain using visual observations, including bird speed and direction, as well as migration intensity (Panuccio et al. 2016b). We investigated the relations between different wind conditions and flight speed, drift compensation, and migration intensity of soaring migrants at the Calabrian side of the Strait of Messina. We hypothesize that in accordance with optimal migration flight theory (Pennycuick 1978; Alerstam 1991), soaring birds will respond to tailwind and crosswinds by changing their air and ground speed and will take advantage of tailwinds to migrate where tailwinds prevail. Furthermore, the birds will try minimizing their drift to reduce deviation from their track. To this end, we tested the following predictions:

- (1) Birds ground speed will increase with increasing tailwind speed (Safi et al. 2013).
- (2) Bird airspeed will decrease with increasing tailwind speed (Pennycuick 1978; Alerstam 1979; Liechti 1995).
- (3) Bird airspeed will increase with increasing crosswind speed (Liechti et al. 1994; Hedenström and Alerstam

Nazionale di Geofisica e Vulcanologia" (Tarquini et al. 2007, 2012). The inset is the location of the study area near the Strait of Messina

2002; Sapir et al. 2014a) in order to minimize their sideways drift (Sapir et al. 2014b).

(4) Tailwind conditions will induce increased migration intensity (Richardson 1978; Gwinner 1990; Fox et al. 2003) such that migration traffic rate over the study area will be higher under tailwind conditions.

Materials and methods

Field data collection

Data collection took place during the spring of 2014 near the edge of a flat highland in the Aspromonte Mountains, Calabria ($38^{\circ} 23'$ N, $15^{\circ} 79'$ E—1030 m a.s.l.), about 7 km inland from the Strait of Messina in southern Italy (Fig. 1). We collected data using a 12-kW, X-band (9.1 GHz) marine radar rotating at 38 RPM with a 2.2-m antenna that was set horizontally with a 22° vertical beam spread and a 1.05° horizontal beam spread. The radar had a detection angle of 240° and a radius of 2 km that was oriented towards the prevalent direction of the incoming migrants (a compass direction of 215°). Experienced birdwatchers carried out daily observations near the radar from sunrise to sunset between April 9th and May 20th, 2014 to characterize species composition and migration traffic rate (Schmidt et al. 2017). The radar operated for 31 days within this period (it was switched off when there was rain or snow) for about 9.5 daylight hours every day (approximately during 8:30-18:00). Radar echoes were annotated with regard to bird species and flock size as identified by the nearby observers (Kerlinger and Gauthreaux 1985a, b). Radar echoes were recorded as 1-Hz video frames and were processed using radR 2.5.1, an open-source program written in the "R" statistical programming language and in "C" code (Taylor et al. 2010). Using radR, we assigned coordinates and time stamps to the radar echoes and created a dataset that we then analyzed using OGIS Desktop 2.8.0 (QGIS Development Team 2015). Subsequent radar echoes (points) of the same target were connected to create a track with a unique ID. Missing points were inserted by interpolation. Afterwards, we calculated the length, mean angular direction from North, and mean ground speed of the tracks. Each track represented one or more birds that were identified as a single target on the radar screen. When the track was produced by more than a single bird, the number of individuals in the flock was registered by the observers. In order to partially mitigate the influence of the movement of local birds, tracks that were directed opposite to the migratory direction were excluded (see "Supplementary Online Materials"). Furthermore, data were inspected and targets consisting of small sized, nonsoaring birds, local breeding birds, and insects (detected as small echoes within 300 m from the radar) were excluded from the dataset. We also considered only tracks with a minimum length of 200 m and a ground speed between 5 m s⁻¹ (= 18 km h⁻¹) and 19.5 m s⁻¹ (= 70 km h⁻¹), to include only large soaring birds (Spaar and Bruderer 1996).

Speed and direction of winds and birds

Wind direction and speed were obtained at 10-min intervals using a weather station with an anemometer positioned 10 m above the ground near the radar station. Wind data were transformed into two biologically meaningful variables, tailwind assistance (TWA) and crosswind (CRW). TWA is the tailwind component (m s⁻¹) of wind velocity or the wind assistance vector in the presumably preferred direction of the bird ("goal", see below), with positive and negative values indicating tailwind and headwinds, respectively (Piersma and Jukema 1990; Åkesson and Hedenström 2000; Sapir et al. 2011). CRW is the sideways component (m s⁻¹) of wind velocity perpendicular to the preferred direction of the bird, with positive and negative values indicating crosswinds coming from the left and the right side of the bird, respectively. TWA and CRW were calculated by the following formulas:

$$TWA = v_W \cdot \cos((\theta_W + 180^\circ) - \theta_b), \tag{1}$$

$$CRW = v_W \cdot \cos((\theta_W + 90^\circ) - \theta_b), \tag{2}$$

where v_W is the measured wind velocity (m s⁻¹), θ_W is wind direction (i.e., the direction from where the wind was blowing) in degrees, and θ_b is the presumed preferred direction of the birds. For a graphic representation of the vectors used in the analyses, see Fig. 2 and Table 1.

We calculated the airspeed (v_a , m s⁻¹) for each track using the following formula (Safi et al. 2013):

$$v_a = \sqrt{\left(v_g - \text{TWA}\right)^2 + \left(\text{CRW}\right)^2},\tag{3}$$

where v_g is the ground speed calculated from the radar track.

For the calculations of TWA (eq. 1) and CRW (eq. 2), one must assess the direction of the presumed destination of the migrants (θ_b). There are several approaches to establish this direction, and we used two of these approaches: one is based on long-distance tracking of birds during their migration after flying through the study area (Sapir et al. 2011), and the other approach is based on the mean direction of bird tracks over the study area (Green 2001; see also Supplementary Materials). We believe that the first approach is preferable because it is



Fig. 2 Graphic representation of the vectors used in the analyses relative to the intended migratory goal (θ_b , light blue arrow). The bird displacement vector is v_g (ground speed, black arrow). The wind vector is v_w (wind speed, red arrow), and it can be decomposed into a forward component (TWA; green arrow) and a side component (CRW; yellow arrow). The airspeed vector is v_a (dashed gray arrow). Units on both axes are in meters per second. For a complete list of symbols and abbreviation, see Table 1

Table 1 Compl	lete list of symb	ols and abbreviations
---------------	-------------------	-----------------------

List of symbols and abbrev	viations
TWA	Tailwind assistance (m s^{-1})
CRW	Crosswind (m s^{-1})
$\mathcal{V}_{\mathcal{W}}$	Wind intensity (m s^{-1})
θ_w	Wind direction (°)
θ_b	Direction of migration goal $(=0^\circ)$
θ_i	Individual direction (°)
v_g	Ground speed (m s^{-1})
v _a	Airspeed (m s^{-1})

based on actual data regarding goal locations or at least part of the routes of specific individuals, which presumably represent the destination of the population of a certain species that passes over a specific area. During spring migration, many Honey buzzards (Pernis apivorus) (about 95% of the birds tracked in this study) arrive to central and northern Europe after passing through Italy (Hake et al. 2003; Meyburg et al. 2011), and consequently, we propose that the intended direction of the passing birds was due north ($\theta_b = 0^\circ$). This information was used in our analyses that are reported in "Results." We nevertheless considered the second approach, which yielded a different estimated goal direction (mean angular direction of the birds recorded by the radar). The analyses that use this estimated goal direction are reported in the Supplementary Materials (Tables S2–S5) and overall agree very well with the findings of the first approach that are reported in "Results."

Following the first approach and using $\theta_b = 0^\circ$ in eqs. (1) and (2), these equations can be simplified to:

$$TWA = v_W \cdot \cos(\theta_W + 180^\circ), \tag{4}$$

and

$$CRW = v_W \cdot \cos(\theta_W + 90^\circ). \tag{5}$$

To annotate bird radar data with the wind data, the time of each track was matched to the nearest 10-min interval at which wind speed and direction were recorded. The distributions of the annotated wind conditions are presented in Fig. S1 together with bird airspeeds and numbers of tracks.

Statistical analysis

We used linear mixed models (LMMs) to analyze the following dependent variables: bird ground speed v_g , airspeed v_a (3), and bird density (the number of bird tracks per hour and per day that were recorded by the radar), in relation to TWA and CRW (independent variables) at daily and hourly timescales (random factors). We followed a top-down strategy for model selection: first, we found the optimal structure of the random part of the model and then of the fixed one (Zuur et al. 2009). Random effects were examined considering models with a random intercept and models with a random intercept and a random slope. Models with a random intercept and those with a random intercept and slope did not differ significantly, so setting the time scale (hour or day) as the random intercept was the most parsimonious model. Afterwards, we found the optimal structure of the fixed component following backward stepwise model selection (starting with all factors and reducing them in the following steps), using the Akaike Information Criterion (AIC; Akaike 1973). Following inspection of model residuals and considering the dispersion of the data, we chose linear mixed models as the most appropriate test (Zuur et al. 2009). We also performed the same analyses for all tracks without any averaging over a certain timescale (hour/day) using simple linear models that were selected by their AIC values.

We used linear models to describe the effects of easterly and westerly crosswinds on ground speed and airspeed. To avoid the use of negative and positive values of CRW to represent the direction of the wind, we additionally used the absolute value of CRW and undertook separate analyses for westerly and easterly winds. Westerly and easterly winds were not equally represented in our sample: 368 tracks were recorded under easterly winds and 8050 tracks under westerly winds, during 22 and 178 h, respectively.

Bird density data were natural log-transformed to achieve normality and homoscedasticity. Statistical analyses were performed with R software version 3.2.2 (R Core Team 2015). LMMs were performed with the "nlme" package (Pinheiro et al. 2016). The mean direction of the tracks, circular standard deviations, and Rayleigh's tests were performed with the "circular" package in R (Agostinelli and Lund 2013).

Data availability The datasets analyzed during the current study are available from the corresponding author on reasonable request.

Results

We analyzed a total of 8418 radar tracks that were recorded through the course of the study (31 days). According to the count data of the nearby observers, 17 species were observed. The Honey buzzard (*Pernis apivorus*) was the most abundant species, accounting for 95.39% of all birds counted, with the second most common species being the Marsh harrier (*Circus aeruginosus*), accounting for 1.52% of the counts (see bird count data in Table S1). The mean (± circular SD) direction of the tracks was $\mu = 32.77^{\circ} \pm 26.65^{\circ}$ with *r* (length of mean vector) = 0.897 (Rayleigh's test, z = 6779.24, P < 0.0001). We tested if the daily bird counts recorded by the observers were correlated with the daily number of tracks recorded by the radar when the radar and the observers worked simultaneously (27 days), and found that bird counts of these two datasets were positively and significantly correlated (Spearman's correlation S = 1828.6, $\rho = 0.44$, P < 0.05).

Bird speed

At the daily scale (averaging over whole days), ground speed increased with CRW, which means that westerly winds increased the birds' ground speed, while easterly winds decreased it (Table 2; Fig. 3a). TWA was not included in the selected model, and therefore, no significant effect of TWA on the birds' ground speed was found at the daily scale. At the hourly scale (averaging over whole hours), ground speed was positively related to both TWA and CRW (Table 2), and therefore, southern and western winds increased bird ground speed while northern and eastern winds decreased it (Fig. 3c, d). The same was found when all the tracks were considered in the analysis without any averaging over time (Table 2). TWA had a negative effect on bird airspeed, such that airspeed decreased with increasing southern winds (tailwinds) and increased with northern winds (headwinds). Furthermore, CRW had a positive effect on bird airspeed such that airspeed increased with westerly winds and decreased with easterly winds at all timescales (Table 3; Fig. 3). Hence, the response of the migrants to easterly and westerly winds was asymmetrical. Under easterly winds, which were overall rare in this area during the course of the study (4.38% of all tracks), the migrants decreased their ground speed (Fig. 4a) and increased their airspeed (Table 4; Fig. 4b). Under westerly winds, the birds slightly increased their ground speed (Fig. 4a) as well as their airspeed (Table 4; Fig. 4b). We note that the response of the birds was much stronger under easterly winds than under westerly winds, as the regression coefficient describing the relationship between Behav Ecol Sociobiol (2018) 72:28

wind speed and bird airspeed was four times higher when birds were flying under easterly winds than under westerly winds (Table 4).

Bird density

Bird density was negatively related to TWA (Table 5) at the daily scale (i.e., the daily number of tracks recorded by the radar) (Fig. 5a) and the same was found at the hourly scale (i.e., number of tracks recorded per hour) (Fig. 5b). In the latter timescale, bird density additionally increased with positive CRW, suggesting that westerly winds induced a more intense migration (Table 5).

Discussion

We found that soaring migrants increased their ground speed and decreased their airspeed when tailwinds prevailed, but unexpectedly, these benefits did not result in more intense migration over the study area, but rather in lower migration traffic rate. Thus, bird response to tailwind seemingly had contrasting aspects, warranting further explanation (see below). We moreover found an asymmetric response of the birds to crosswind, compensating when winds blew towards the sea and drifting when winds blew towards land. We propose that the response of the birds to wind conditions is modulated by the geography of their migration route at two spatial scales. The spatial scales we refer to are the local scale of the study area and its surroundings, and the regional scale that includes the south-Italian peninsula and the sea around it. We propose that the atmospheric and geographical conditions experienced at the local scale can affect the migration route at the regional scale and that regional scale dynamics of migrants may also affect local scale patterns. Our results suggest that migrating

 Table 2
 Final models of the relationships between bird ground speed (dependent variable) and tailwind and crosswind speed and their interaction at hourly and daily timescales

Time scale	Model type	Explanatory variable	Estimate	SE	t	Р
Daily	Linear mixed model	(Intercept)	13.95	0.19	73.6	< 0.0001
		CRW	0.13	0.02	6.2	< 0.0001
Hourly	Linear mixed model	(Intercept)	13.72	0.13	106.2	< 0.0001
		TWA	0.20	0.03	6.9	< 0.0001
		CRW	0.17	0.03	6.5	< 0.0001
		$TWA \times CRW$	-0.05	0.01	- 5.9	< 0.0001
No averaging (all tracks)	Linear model	(Intercept)	13.68	0.08	164.8	< 0.0001
$(R^2 = 0.04)$		TWA	0.18	0.03	6.2	< 0.0001
		CRW	0.18	0.03	6.9	< 0.0001
		$TWA \times CRW$	-0.06	0.01	- 6.1	< 0.0001

TWA tailwind speed in m s⁻¹, *CRW* crosswind speed in m s⁻¹ (see "Materials and Methods" for additional details)



Fig. 3 Average bird speeds fitted with linear regression lines (colored areas 95% CI) in relation to crosswind speed (CRW **a**, **c**) and tailwind speed (TWA **b**, **d**) at daily and hourly timescales

raptors modulate different aspects of their migration properties with regard to both wind and geography (see also Mandel et al. 2008; Mellone et al. 2011; Nilsson et al. 2014; Åkesson et al. 2016). Specifically, we show that soaring migrants are able to cover more distance (higher ground speed) under tailwind conditions while lowering their airspeed, and compensate for potential dangerous oversea drift when pushed by easterly winds but not when drifting by westerlies over land.

The response of the birds affects their speed, which may eventually influence their time of arrival to breeding grounds (Gordo 2007), breeding timing, and reproductive output (Saino et al., 2004a, b; Szép et al. 2006; Gordo and Sanz 2008). Consequently, the ability of migrating birds to

Table 3 Final models of therelationships between airspeed(dependent variable) and tailwindand crosswind speed and theirinteraction at daily and hourlytimescales

Time scale	Model type	Explanatory variable	Estimate	SE	t	Р
Daily	Linear mixed	(Intercept)	13.90	0.19	72.9	< 0.0001
	model	TWA	-1.02	0.03	-34.1	< 0.0001
		CRW	0.26	0.03	9.8	< 0.0001
		$TWA \times CRW$	-0.88	0.02	0.7	0.5059
Hourly	Linear mixed model	(Intercept)	13.70	0.13	105.3	< 0.0001
		TWA	-0.81	0.03	-28.1	< 0.0001
		CRW	0.30	0.03	12.0	< 0.0001
		$TWA \times CRW$	-0.04	0.01	-4.5	< 0.0001
No averaging (all tracks) $(R^2 = 0.36)$	Linear model	(Intercept)	13.65	0.08	167.9	< 0.0001
		TWA	-0.84	0.03	-28.9	< 0.0001
		CRW	0.31	0.03	12.5	< 0.0001
		$TWA\timesCRW$	-0.04	0.01	-4.8	< 0.0001

TWA tailwind speed in m s⁻¹, CRW crosswind speed in m s⁻¹ (see "Materials and Methods" for additional details)



Fig. 4 Hourly averaged ground speed (a) and airspeed (b) fitted with linear regression lines (colored areas 95% CI) in relation to easterly and westerly crosswinds. CRW absolute crosswind speed

successfully modulate their flight properties with regard to different environmental conditions can have far-reaching consequences beyond individual fitness, for example for their population dynamics (Alves et al. 2013; Flack et al. 2016). For soaring migrants, and specifically for Honey buzzards that constitute the vast majority of the soaring migrants in this area, successful negotiation of their movements under variable meteorological conditions is critical for accomplishing their migration in timely manner that will allow them to successfully breed in their highly seasonal breeding habitats (Thorup et al. 2003; Vansteelant et al. 2017a, b). Our findings suggest that migrating Honey buzzards (Pernis apivorus) integrate information regarding the highly dynamic meteorological conditions they encounter during their travel with the geographical features of the flyway to take advantage of profitable conditions and avoid risks associated with drifting over the sea during unfavorable weather conditions (Mote 1969; Kerlinger 1984, 1985; Thorup et al. 2003; Agostini et al. 2016).

We found that soaring migrants adjusted their ground speed and airspeed with respect to wind speed and direction. Our results regarding the relationships between tailwind support and bird airspeed and ground speed confirmed prediction 1, which states that birds are expected to achieve higher ground speed with tailwinds, and prediction 2, which states that bird airspeed is expected to decrease with tailwind. In fact, as far as prediction 1 is concerned, the ground speed of soaring migrants (primarily Honey buzzards in this case) increased with tailwind at the hourly scale, as found for many other flying species of birds, bats, and insects (Srygley and Dudley 2008; Safi et al. 2013; Sapir et al. 2014a; Shamoun-Baranes et al. 2017), and specifically in soaring raptors (Spaar and Bruderer 1996; Mellone et al. 2015). A recent study found that Montagu's harriers (Circus pygargus), which use both soaring and flapping flight during migration, increased their daily flight duration and distance when under tailwind conditions and interrupted their flight under headwinds (Klaassen et al. 2017). Honey buzzards select tailwinds to travel faster and further as well (Vansteelant et al. 2015; Panuccio et al. 2016a), and while doing so, the birds can move faster through geographical barriers such as water bodies (Panuccio et al. 2016a). At the daily scale, we did not find the same effect, probably because of the variability in wind conditions over this longer time period, and the averaging of the different responses of the migrants to these variable conditions. The significant response at the hourly scale and the non-

Table 4 Linear models with ground speed and airspeed as dependent variables and easterly and westerly wind speeds as explanatory variables

	Ground spe	Ground speed			Airspeed	Airspeed				
	Estimate	SE	t	Р	R^2	Estimate	SE	t	Р	R^2
(Intercept) Easterly CRW	14.74 - 1.03	0.27 0.13	55.5 - 7.9	< 0.0001 < 0.0001	0.14	11.80 1.24	0.30 0.15	39.5 8.3	< 0.0001 < 0.0001	0.16
(Intercept) Westerly CRW	13.28 0.30	0.07 0.02	188.1 13.7	< 0.0001 < 0.0001	0.02	15.53 0.28	0.08 0.03	183.9 10.6	< 0.0001 < 0.0001	0.01

CRW crosswind speed in m s⁻¹ (see "Materials and Methods" for additional details)

 Table 5
 Final models of the relationships between number of tracks (transformed with natural logarithm) as dependent variable and tailwind and crosswind speed and their interaction at daily and hourly timescales

Time scale	Explanatory variable	Estimate	SE	t	Р
Daily	(Intercept)	4.50	0.27	16.67	< 0.0001
	TWA	-0.43	0.14	-3.07	< 0.01
Hourly	(Intercept)	2.24	0.18	12.40	< 0.0001
	TWA	-0.22	0.06	-3.91	< 0.001
	CRW	0.15	0.05	2.77	< 0.01

TWA tailwind speed in m s⁻¹, *CRW* crosswind speed in m s⁻¹ (see "Materials and Methods" for additional details)

significant response at the daily scale may suggest that the temporal scale at which the birds actually sense, and respond to, wind conditions is more detailed than the daily scale (Nathan et al. 2005). As for prediction 2 that was supported by our results, optimal flight theory for flapping migrants portrays that the reason for the decrease of airspeed with increasing tailwind speed is likely due to energy saving during migration (Pennycuick 1978; Liechti et al. 1994; Liechti 1995; Safi et al. 2013). For soaring migrants, reducing airspeed under tailwinds allows the birds to attain low sink rate and by that to cover larger distances while decreasing the risk of reaching the ground or switching to energy-expensive flapping flight (Spaar and Bruderer 1996; Horvitz et al. 2014; Harel et al. 2016).

As for prediction 3, theoretical calculations suggest that birds should increase their airspeed in response to crosswind in order to reduce their lateral drift or to fully compensate for it (Liechti et al. 1994). This response has been so far documented only in fruit bats (Sapir et al. 2014a). Bird response to crosswind over the study area fitted only partially to prediction 3. In fact, ground speed increased in relation to crosswind from the west and decreased under easterly crosswind, and bird response to crosswind was asymmetrical in relation to crosswind direction (from the east or from the west), depending on the expected outcome of the potential drift. This may suggest that birds employed an adaptive drift when westerly winds that advanced the birds over land in Calabria prevailed, while compensating for drift when easterly winds blew and may consequently carry the birds over the sea. This compensation was found under headwinds, as well as under tailwinds (see below). Avoiding flight over the sea may allow the birds to keep low metabolic rates using soaring flight (Baudinette and Schmidt-Nielsen 1974; Bevan et al. 1995; Sapir et al. 2010) and may further lower mortality risks en route (Thorup et al. 2003). We note that a recent study demonstrated an asymmetric response of nocturnally migrating songbirds to crosswind near the North American Atlantic coast in which the birds drifted when flying over inland areas, but compensated for drift to avoid flying over the ocean near the coast (Horton et al. 2016). The reason for this response of nocturnally or early-morning migrating passerines to crosswinds near coasts might be due to increased predation risks when the nocturnal flight extends to daytime hours (Bourne 1980) and possible dehydration (Klaassen 2004). Yet, exhaustion due to enduring flight over water and eventually mortality caused by drowning may constitute a strong selection force for both diurnal soaring birds and nocturnal flapping passerines during migration. Yet, flight over the sea may still be



Fig. 5 The number of radar tracks in relation to wind speed at daily (**a**) and hourly (**b**) timescales fitted with linear regression lines (colored areas 95% CI). TWA tailwind speed in m s⁻¹, CRW crosswind speed in m s⁻¹

advantageous when considering an alternative long detour over land in some geographic areas, especially when assisting winds may facilitate a rapid cross sea migration (Alerstam 2001; see below). Why birds experiencing tailwind conditions did not drift over the sea could relate to different factors. For example, these individuals were more risk-averse and avoided flying over the sea whenever possible, and migrate through the area when most other birds avoided flying through it (see below). Also, birds may avoid flying over the sea when their physiological state is deteriorated and may not allow them to undertake an enduring cross sea flight.

Surprisingly, soaring birds migrated over the study area in greater numbers under headwind conditions, in contrast with prediction 4 that predicted increased migration intensity with tailwinds, as has been found in numerous studies of migrating animals (Richardson 1978; Perdeck and Speek 1984; Gwinner 1990; Erni et al. 2002; Van Belle et al. 2007; Alerstam et al. 2011; Hu et al. 2016). Migrants are expected to avoid headwinds in order to migrate more quickly and spend less energy during flight (Alerstam and Hedenstrom 1998; Klaassen et al. 2017; Vansteelant et al. 2017b). The apparent advantageous assistance of tailwind to the birds' overall flight speed (as demonstrated by their high ground speed under tailwinds) seemed to be ignored by the birds. We suggest that a likely reason for this counterintuitive pattern is probably sea crossing directly from the Sicilian side of the Strait of Messina or from the Calabrian coast (west of the study area) when tailwinds prevailed in the entire region, including, but not only, in the study area (Agostini and Panuccio 2005; Panuccio 2011; Agostini et al. 2015, 2016). These flights, nevertheless, were not recorded as they took place beyond the range of the radar. We propose that the spatial scale of wind assistance patterns possibly extended over tens or even several hundreds of kilometers, allowing the birds to undertake a long cross sea journey and eventually causing a local decline in migration intensity in the study area under these conditions. Hence, wind assistance affected route selection of Honey buzzards and likely other soaring migrants passing over Sicily and Calabria such that they did not fly over land following the coastline but rather flew over the sea. This is expected when the cost of transport while flying over the sea is overall lower than that of the overland detour (Meyer et al. 2000; Alerstam 2001; Agostini et al. 2015; Nourani et al. 2016). A straight sea crossing from the Strait of Messina towards the north (about 200 km before reaching the mainland) would allow soaring birds to reduce migration distance by about 80 km compared to a detour over the land. Wind and thermal conditions affected migratory directions in autumn soaring migrants that crossed the Strait of Gibraltar: sea crossing occurred in favorable following and easterly winds, but no sea crossings were observed in unfavorable westerly winds and headwinds. Under the latter wind conditions, migration was carried out above land through a long detour until the

birds reached the shortest crossing point in the Strait of Gibraltar (Meyer et al. 2000). Panuccio et al. (2016a) suggested that Honey buzzards crossing the Aegean Sea selected days with strong tailwinds, presumably increasing the bird's ground speed and decreasing the total flight duration while passing through this area. We note that unlike flapping migrants such as songbirds, waterfowl, and waders that primarily consider their cost of transport in relation to the added mass of the fuel they carry during their travel (Alerstam 2001), soaring migrants may additionally consider the costs and risks of flying using flapping flight over the sea as opposed to their regular soaring flight over land. To date, there are no explicit theoretical formulations that predict when overland detours versus cross sea shortcuts by soaring migrants are expected, as has been done for flapping flyers by Alerstam (2001).

Our study highlights the complexity of the dynamic aerial environment and changing geographic features throughout the migrants' travel route, requiring high behavioral plasticity of their flight properties to properly respond to the spatiotemporal dynamics of their aerial environment. Cross taxa comparisons and studies undertaken in different geographic areas may help assessing the generality of our findings. It is furthermore important to address meteorological and geographic effects on the migrants' cost of transport throughout the entire migration routes in order to deduce energy consumption considerations, route choice, and carry-over effects that may substantially affect animal fitness.

Acknowledgements We thank Viviana Stanzione for her help during the fieldwork and David Troupin for his help with the data analysis. We would like to thank two reviewers that substantially improved the manuscript with their comments and suggestions.

Funding Data collection was supported by Terna Rete Italia S.p.A. Ornis italica supported part of the fieldwork. PB was partially funded by the Minerva Center for Movement Ecology, and by COST-European Cooperation in Science and Technology through the Action ES1305 "European Network for the Radar Surveillance of Animal Movement" (ENRAM).

Compliance with ethical standards

Ethical approval This study did not require any ethical approval.

Conflict of interest The authors declare that they have no conflict of interest.

References

Agostinelli C, Lund U (2013) R package "circular": circular statistics. R package version 0.4–7, https://r-forge.r-project.org/projects/circular/

Agostini N, Gustin M, von Hardenberg J, Panuccio M (2016) Wind patterns affect migration flyways and flock size of a soaring bird over sea. Avian Biol Res 9(3):1-8. https://doi.org/10.3184/ 175815516X14627928448105

- Agostini N, Panuccio M (2005) Analysis of the spatial migration patterns of adult Honey buzzards (*Pernis apivorus*) during spring and autumn in the Central Mediterranean. Ring 27(2):215–220. https://doi.org/10.2478/v10050-008-0018-x
- Agostini N, Panuccio M, Pasquaretta C (2015) Morphology, flight performance, and water crossing tendencies of Afro-Palearctic raptors during migration. Curr Zool 61(6):951–958. https://doi.org/10. 1093/czoolo/61.6.951
- Akaike H (1973) Information theory and an extension of the maximum likelihood principle. In: 2nd International Symposium on Information Theory Akadémiai Kiadó, Budapest, pp 267–281
- Åkesson S, Bianco G, Hedenström A (2016) Negotiating an ecological barrier: crossing the Sahara in relation to winds by common swifts. Philos T Roy Soc B 371(1704):20150393. https://doi.org/10.1098/ rstb.2015.0393
- Åkesson S, Hedenström A (2000) Wind selectivity of migratory flight departures in birds. Behav Ecol Sociobiol 47(3):140–144. https://doi.org/10.1007/s002650050004
- Alerstam T (1979) Wind as selective agent in bird migration. Ornis Scand 10(1):76–93. https://doi.org/10.2307/3676347
- Alerstam T (2001) Detours in bird migration. J Theor Biol 209(3):319– 331. https://doi.org/10.1006/jtbi.2001.2266
- Alerstam T (1991) Bird flight and optimal migration. Trends Ecol Evol 6: 210–215. https://doi.org/10.1016/0169-5347(91)90024-R
- Alerstam T, Chapman J, Bäckman J, Smith AD, Karlsson H, Nilsson C, Reynolds DR, Klaassen RHG, Hill JK (2011) Convergent patterns of long-distance nocturnal migration in noctuid moths and passerine birds. Proc R Soc Lond B 278(1721):3074–3080. https://doi.org/10.1098/rspb.20
- Alerstam T, Hedenstrom A (1998) The development of bird migration theory. J Avian Biol 29(4):343–369. https://doi.org/10.2307/ 3677155
- Alerstam T, Pettersson SG (1977) Why do migrating birds fly along coastlines? J Theor Biol 65(4):699–712. https://doi.org/10.1016/ 0022-5193(77)90016-9
- Alves J, Gunnarsson TG, Hayhow DB, Appleton GF, Potts PM, Sutherland WJ, Gill JA (2013) Costs, benefits, and fitness consequences of different migratory strategies. Ecology 94(1):11–17. https://doi.org/10.1890/12-0737.1
- Baudinette RV, Schmidt-Nielsen K (1974) Energy cost of gliding flight in herring gulls. Nature 248(5443):83–84. https://doi.org/10.1038/ 248083b0
- Bensusan KJ, Garcia EFJ, Cortes JE (2007) Trends in abundance of migrating raptors at Gibraltar in spring. Ardea 95(1):83–90. https://doi.org/10.5253/078.095.0109
- Bevan RM, Butler PJ, Woakes AJ, Prince PA (1995) The energy expenditure of free-ranging black-browed albatrosses. Philos T Roy Soc B 350(1332):119–131. https://doi.org/10.1098/rstb.1995.0146
- Bildstein KL, Bechard MJ, Farmer C, Newcomb L (2009) Narrow sea crossings present major obstacles to migrating Griffon Vultures *Gyps fulvus*. Ibis 151(2):382–391. https://doi.org/10.1111/j.1474-919X.2009.00919.x
- Boano G, Brichetti P, Foschi UF (2010) "La Niña"-driven Atlantic storms affect winter survival of Mediterranean Cory's Shearwaters. Ital J Zool 77(4):460–468. https://doi.org/10.1080/11250000903469017
- Bohrer G, Brandes D, Mandel JT, Bildstein KL, Miller TA, Lanzone M, Katzner T, Maisonneuve C, Tremblay JA (2012) Estimating updraft velocity components over large spatial scales: contrasting migration strategies of golden eagles and turkey vultures. Ecol Lett 15(2):96– 103. https://doi.org/10.1111/j.1461-0248.2011.01713.x
- Bourne WRP (1980) The midnight descent, dawn ascent and reorientation of land birds migrating across the North Sea in autumn. Ibis 122(4):536–540. https://doi.org/10.1111/j.1474-919X.1980. tb00915.x

- Erni B, Liechti F, Underhill LG, Bruderer B (2002) Wind and rain govern the intensity of nocturnal bird migration in central Europe—a loglinear regression analysis. Ardea 90:155–166
- Fayet AL, Freeman R, Shoji A, Padget O, Perrins CM, Guilford T (2015) Lower foraging efficiency in immatures drives spatial segregation with breeding adults in a long-lived pelagic seabird. Anim Behav 110:79–89. https://doi.org/10.1016/j.anbehav.2015.09.008
- Flack A, Fiedler W, Blas J et al (2016) Costs of migratory decisions: a comparison across eight white stork populations. Sci Adv 2(1): e1500931. https://doi.org/10.1126/sciadv.1500931
- Fox AD, Glahder CM, Walsh AJ (2003) Spring migration routes and timing of Greenland white-fronted geese—results from satellite telemetry. Oikos 103(2):415–425. https://doi.org/10.1034/j.1600-0706.2003.12114.x
- Gordo O (2007) Why are bird migration dates shifting? A review of weather and climate effects on avian migratory phenology. Clim Res 35:37–58. https://doi.org/10.3354/cr00713
- Gordo O, Sanz JJ (2008) The relative importance of conditions in wintering and passage areas on spring arrival dates: the case of long-distance Iberian migrants. J Ornithol 149(2):199–210. https://doi.org/10.1007/s10336-007-0260-z
- Green M (2001) Is wind drift in migrating barnacle and brent geese, Branta leucopsis and Branta bernicla, adaptive or non-adaptive? Behav Ecol Sociobiol 50(1):45–54. https://doi.org/10.1007/ s002650100333
- Green M, Alerstam T, Clausen P, Drent R, Ebbinge BS (2002) Darkbellied Brent Geese *Branta bernicla bernicla*, as recorded by satellite telemetry, do not minimize flight distance during spring migration. Ibis 144(1):106–121. https://doi.org/10.1046/j.0019-1019. 2001.00017.x
- Gwinner E (1990) Introduction. In: Gwinner E (ed) Bird migration: physiology and ecophysiology. Springer-Verlag, Berlin, pp 1–4. https://doi.org/10.1007/978-3-642-74542-3 1
- Hake M, Kjellen N, Alerstam T (2003) Age-dependent migration strategy in honey buzzards *Pernis apivorus* tracked by satellite. Oikos 103: 385–396
- Harel R, Getz W, Nathan R et al (2016) Decision-making by a soaring bird: time, energy and risk considerations at different spatiotemporal scales. Philos T Roy Soc B 371(1704):20150397. https://doi.org/10.1098/rstb.2015.0397
- Hedenström A, Alerstam T (2002) Adaptive variation of airspeed in relation to wind, altitude and climb rate by migrating birds in the Arctic. Behav Ecol Sociobiol 52(4):308–317. https://doi.org/10. 1007/s00265-002-0504-0
- Horton KG, Van Doren BM, Stepanian PM, Hochachka WM, Farnsworth A, Kelly JF (2016) Nocturnally migrating songbirds drift when they can and compensate when they must. Sci Rep 6(1):21249. https://doi.org/10.1038/srep21249
- Horvitz N, Sapir N, Liechti F, Avissar R, Mahrer I, Nathan R (2014) The gliding speed of migrating birds: slow and safe or fast and risky? Ecol Lett 17(6):670–679. https://doi.org/10.1111/ele.12268
- Hu G, Reynolds DR, Sapir N, Chapman JW (2016) Mass seasonal bioflows of high-flying insect migrants. Science 354(6319):1584– 1587. https://doi.org/10.1126/science.aah4379
- Kerlinger P (1985) Water-crossing behavior of raptors during migration. Wilson Bull 97:109–113
- Kerlinger P (1984) Flight behaviour of Sharp-shinned Hawks during migration. 2: over water. Anim Behav 32(4):1029–1034. https://doi.org/10.1016/S0003-3472(84)80217-1
- Kerlinger P, Gauthreaux SA (1985a) Flight behavior of raptors during spring migration in South Texas studied with radar and visual observations. J Field Ornithol 56:394–402
- Kerlinger P, Gauthreaux SA (1985b) Seasonal timing, geographic distribution, and flight behavior of broad-winged hawks during spring migration in South Texas: a radar and visual study. Auk 102:735–743

- Kjellén N, Roos G (2000) Population trends in Swedish raptors demonstrated by migration counts at Falsterbo, Sweden 1942–97. Bird Study 47(2):195–211. https://doi.org/10.1080/00063650009461175
- Klaassen M (2004) May dehydration risk govern long-distance migratory behaviour? J Avian Biol 35:4–6
- Klaassen M, Beekman JH, Kontiokorpi J, Mulder RJW, Nolet BA (2004) Migrating swans profit from favourable changes in wind conditions at low altitude. J Ornithol 145(2):142–151. https://doi.org/10.1007/ s10336-004-0025-x
- Klaassen RHG, Hake M, Strandberg R, Alerstam T (2011) Geographical and temporal flexibility in the response to crosswinds by migrating raptors. Proc R Soc Lond B 278(1710):1339–1346. https://doi.org/ 10.1098/rspb.2010.2106
- Klaassen RHG, Schlaich AE, Bouten W, Koks BJ (2017) Migrating Montagu's harriers frequently interrupt daily flights in both Europe and Africa. J Avian Biol 48(1):180–190. https://doi.org/10.1111/jav. 01362
- La Sorte FA, Fink D, Hochachka WM, Kelling S (2016) Convergence of broad-scale migration strategies in terrestrial birds. Proc R Soc B 283(1823):20152588. https://doi.org/10.1098/rspb.2015.2588
- Liechti F (2006) Birds: blowin' by the wind? J Ornithol 147(2):202–211. https://doi.org/10.1007/s10336-006-0061-9
- Liechti F (1995) Modelling optimal heading and airspeed of migrating birds in relation to energy expenditure and wind influence. J Avian Biol 26(4):330–336. https://doi.org/10.2307/3677049
- Liechti F, Hedenström A, Alerstam T (1994) Effects of sidewinds on optimal flight speed of birds. J Theor Biol 170:219–225
- Ma Z, Hua N, Zhang X, Guo H, Zhao B, Ma Q, Xue W, Tang C (2011) Wind conditions affect stopover decisions and fuel stores of shorebirds migrating through the south Yellow Sea. Ibis 153(4):755–767. https://doi.org/10.1111/j.1474-919X.2011.01164.x
- Malmiga G, Nilsson C, Bäckman J, Alerstam T (2014) Interspecific comparison of the flight performance between sparrowhawks and common buzzards migrating at the Falsterbo peninsula : a radar study. Curr Zool 60(5):670–679. https://doi.org/10.1093/czoolo/60.5.670
- Mandel JT, Bildstein KL, Bohrer G, Winkler DW (2008) Movement ecology of migration in turkey vultures. P Natl Acad Sci USA 105(49):19102–19107. https://doi.org/10.1073/pnas.0801789105
- Mellone U, De La Puente J, López-López P, Limiñana R, Bermejo A, Urios V (2015) Seasonal differences in migration patterns of a soaring bird in relation to environmental conditions: a multi-scale approach. Behav Ecol Sociobiol 69(1):75–82. https://doi.org/10.1007/ s00265-014-1818-4
- Mellone U, Klaassen RHG, García-Ripollés C, Limiñana R, López-López P, Pavón D, Strandberg R, Urios V, Vardakis M, Alerstam T (2012) Interspecific comparison of the performance of soaring migrants in relation to morphology, meteorological conditions and migration strategies. PLoS One 7(7):e39833. https://doi.org/10. 1371/journal.pone.0039833
- Mellone U, López-López P, Limiñana R, Urios V (2011) Weather conditions promote route flexibility during open ocean crossing in a long-distance migratory raptor. Int J Biometeorol 55(4):463– 468. https://doi.org/10.1007/s00484-010-0368-3
- Meyburg B, Ziesemer F, Martens HD, Meyburg C (2011) On the biology of the Honey Buzzard (*Pernis apivorus*)—results of satellite tracking. 8th Conference of the European Ornithologists' Union, Riga, Latvia
- Meyer SK, Spaar R, Bruderer B (2000) To cross the sea or to follow the coast? Flight directions and behaviour of migrating raptors approaching the Mediterranean Sea in autumn. Behaviour 137: 379–399

Mote WR (1969) Turkey Vultures land on vessel in fog. Auk 86:766-767

Nathan R, Sapir N, Trakhtenbrot A et al (2005) Long-distance biological transport processes through the air: can nature's complexity be unfolded in silico? Divers Distrib 11(2):131–137. https://doi.org/10.1111/j.1366-9516.2005.00146.x

- Nilsson C, Bäckman J, Alerstam T (2014) Are flight paths of nocturnal songbird migrants influenced by local coastlines at a peninsula? Curr Zool 60(5):660–669. https://doi.org/10.1093/czoolo/60.5.660
- Nourani E, Yamaguchi NM, Manda A, Higuchi H (2016) Wind conditions facilitate the seasonal water-crossing behaviour of Oriental Honey-buzzards *Pernis ptilorhynchus* over the East China Sea. Ibis 158(3):506–518. https://doi.org/10.1111/ibi.12383
- Panuccio M (2011) Wind effects on visible raptor migration in spring at the Strait of Messina, southern Italy. J Raptor Res 45(1):88–92. https://doi.org/10.3356/JRR-10-16.1
- Panuccio M, Agostini N, Premuda G (2012) Ecological barriers promote risk minimisation and social learning in migrating short-toed snake eagles. Ethol Ecol Evol 24(1):74–80. https://doi.org/10.1080/ 03949370.2011.583692
- Panuccio M, Barboutis C, Chiatante G, Evangelidis A, Agostini N (2016a) Pushed by increasing air temperature and tailwind speed: weather selectivity of raptors migrating across the Aegean Sea. Ornis Fenn 93:1–13
- Panuccio M, Stanzione V, Catoni C, Santini M, Dell'Omo G (2016b) Radar tracking reveals influence of crosswinds and topography on migratory behavior of European honey buzzards. J Ethol 34(1):73– 77. https://doi.org/10.1007/s10164-015-0448-2
- Pennycuick CJ (1978) Fifteen testable predictions about bird flight. Oikos 30(2):165–176. https://doi.org/10.2307/3543476
- Perdeck AC, Speek G (1984) A radar study on the influence of expected ground speed, cloudiness and temperature on diurnal migration intensity. Ardea 72:189–198
- Piersma T, Jukema J (1990) Budgeting the flight of a long-distance migrant: changes in nutrient reserve levels of bar-tailed godwits at successive spring staging sites. Ardea 78:315–337
- Pinheiro J, Bates D, DebRoy S, Sarkar D, Heisterkamp S, Van Willigen B (2016) nlme: linear and nonlinear mixed effects models. R Packag Version 3:1–127 http://CRAN.R-project.org/package=nlme
- QGIS Development Team (2015) QGIS Geographic Information System, http://qgis.osgeo.org
- Core Team R (2015) R: a language and environment for statistical computing. R Foundation for Statistical Computing, Vienna http:// www.R-project.org
- Richardson WJ (1978) Timing and amount of bird migration in relation to weather: a review. Oikos 30(2):224–272. https://doi.org/10.2307/3543482
- Rotics S, Kaatz M, Resheff YS et al (2016) The challenges of the first migration: movement and behavior of juvenile versus adult white storks with insights regarding juvenile mortality. J Anim Ecol 85(4): 938–947. https://doi.org/10.1111/1365-2656.12525
- Safi K, Kranstauber B, Weinzierl R et al (2013) Flying with the wind: scale dependency of speed and direction measurements in modelling wind support in avian flight. Mov Ecol 1:4
- Saino N, Szep T, Ambrosini R, Romano M, Møller AP (2004a) Ecological conditions during winter affect sexual selection and breeding in a migratory bird. Proc R Soc Lond B 271(1540):681– 686. https://doi.org/10.1098/rspb.2003.2656
- Saino N, Szép T, Romano M, Rubolini D, Spina F, Møller AP (2004b) Ecological conditions during winter predict arrival date at the breeding quarters in a trans-Saharan migratory bird. Ecol Lett 7(1):21–25. https://doi.org/10.1046/j.1461-0248.2003.00553.x
- Sapir N, Horvitz N, DKN D, Fahr J, Wikelski M (2014a) Commuting fruit bats beneficially modulate their flight in relation to wind. Proc R Soc B 281(1782):20140018. https://doi.org/10.1098/rspb.2014. 0018
- Sapir N, Horvitz N, Wikelski M, Avissar R, Nathan R (2014b) Compensation for lateral drift due to crosswind in migrating European Bee-eaters. J Ornithol 155(3):745–753. https://doi.org/ 10.1007/s10336-014-1060-x
- Sapir N, Wikelski M, Avissar R, Nathan R (2011) Timing and flight mode of departure in migrating European bee-eaters in relation to multi-

scale meteorological processes. Behav Ecol Sociobiol 65(7):1353–1365. https://doi.org/10.1007/s00265-011-1146-x

- Sapir N, Wikelski M, MD MC, Pinshow B, Nathan R (2010) Flight modes in migrating European Bee-eaters: heart rate may indicate low metabolic rate during soaring and gliding. PLoS One 5(11): e13956. https://doi.org/10.1371/journal.pone.0013956
- Schmidt M, Aschwanden J, Liechti F, Wichmann G, Nemeth E (2017) Comparison of visual bird migration counts with radar estimates. Ibis 159(3):491–497. https://doi.org/10.1111/ibi.12473
- Shamoun-Baranes J, Felix B, Wouter L (2017) Atmospheric conditions create freeways, detours and tailbacks for migrating birds. J Comp Physiol A 203(6-7):509–529. https://doi.org/10.1007/s00359-017-1181-9
- Spaar R, Bruderer B (1996) Soaring migration of steppe eagles *Aquila nipalensis* in Southern Israel: flight behaviour under various wind and thermal conditions. J Avian Biol 27(4):289–301. https://doi.org/10.2307/3677260
- Spaar R, Bruderer B (1997) Optimal flight behavior of soaring migrants: a case study of migrating steppe buzzards, *Buteo buteo vulpinus*. Behav Ecol 8(3):288–297. https://doi.org/10.1093/beheco/8.3.288
- Srygley RB, Dudley R (2008) Optimal strategies for insects migrating in the flight boundary layer: mechanisms and consequences. Integr Comp Biol 48(1):119–133. https://doi.org/10.1093/icb/icn011
- Szép T, Møller AP, Piper S, Nuttall R, Szabó ZD, Pap PL (2006) Searching for potential wintering and migration areas of a Danish Barn Swallow population in South Africa by correlating NDVI with survival estimates. J Ornithol 147(2):245–253. https://doi.org/10. 1007/s10336-006-0060-x
- Tarquini S, Isola I, Favalli M, Mazzarini F, Bisson M, Pareschi MT, Boschi E (2007) TINITALY/01: a new triangular irregular network of Italy. Ann Geophys-Italy 50:407–425
- Tarquini S, Vinci S, Favalli M, Doumaz F, Fornaciai A, Nannipieri L (2012) Release of a 10-m-resolution DEM for the Italian territory: comparison with global-coverage DEMs and anaglyph-

mode exploration via the web. Comput Geosci 38(1):168–170. https://doi.org/10.1016/j.cageo.2011.04.018

- Taylor PD, Brzustowski JM, Matkovich C, Peckford ML, Wilson D (2010) radR: an open-source platform for acquiring and analysing data on biological targets observed by surveillance radar. BMC Ecol 10(1):22. https://doi.org/10.1186/1472-6785-10-22
- Thorup K, Alerstam T, Hake M, Kjellén N (2003) Bird orientation: compensation for wind drift in migrating raptors is age dependent. Proc R Soc Lond B 270(Suppl_1):S8–S11. https://doi.org/10.1098/rsbl. 2003.0014
- Van Belle J, Shamoun-Baranes J, Van Loon E, Bouten W (2007) An operational model predicting autumn bird migration intensities for flight safety. J Appl Ecol 44(4):864–874. https://doi.org/10.1111/j. 1365-2664.2007.01322.x
- Vansteelant WMG, Bouten W, Klaassen RHG, Koks BJ, Schlaich AE, van Diermen J, van Loon EE, Shamoun-Baranes J (2015) Regional and seasonal flight speeds of soaring migrants and the role of weather conditions at hourly and daily scales. J Avian Biol 46(1):25–39. https://doi.org/10.1111/jav.00457
- Vansteelant WMG, Shamoun-baranes J, van Manen W, van Diermen J, Bouten W (2017a) Seasonal detours by soaring migrants shaped by wind regimes along the East Atlantic Flyway. J Anim Ecol 86(2): 179–191. https://doi.org/10.1111/1365-2656.12593
- Vansteelant WMG, Shamoun-baranes J, Mclaren J, van Diermen J, Bouten W (2017b) Soaring across continents : decision-making of a soaring migrant under changing atmospheric conditions along an entire flyway. J Avian Biol 48(6):887–896. https://doi.org/10.1111/ jav.01298
- Vidal-Mateo J, Mellone U, López-López P, De La Puente J, García-Ripollés C, Bermejo A, Urios V (2016) Wind effects on the migration routes of trans-Saharan soaring raptors: geographical, seasonal and interspecific variation. Curr Zool 62:89–97
- Zuur AF, Ieno EN, Walker N, Saveliev AA, Smith GM (2009) Mixed effects models and extension in ecology with R. Springer, New York. https://doi.org/10.1007/978-0-387-87458-6